Author's response to reviews

Title: A prospective evaluation of the safety and efficacy of the TAXUS element paclitaxel-eluting coronary stent system for the treatment of de novo coronary artery lesions: Design and statistical methods of the PERSEUS clinical program

Authors:

Dominic J Allocco (Dominic.allocco@bsci.com)
Louis A Cannon (lcannon@cvrcnm.com)
Amy Britt (Amy.britt@bsci.com)
John E Heil (John.heil@bsci.com)
Andrey Nersesov (Andrey.nersesov@bsci.com)
Scott Wehrenberg (Scott.wehrenberg@bsci.com)
Keith D Dawkins (Keith.dawkins@bsci.com)
Dean J Kereiakes (lindner@fuse.net)

Version: 5 Date: 6 January 2010

Author's response to reviews:

Minor proofreading comments have been inserted as tracked changes. Note that higher resolution versions of Appendix B Figure 2 and Appendix B Figure 3 are available (if needed) and will be sent under separate cover via email to indiepreaccept@biomedcentral.com as they were too large to upload (kept getting system timeout errors).